Author's response to reviews

Title: Ethnicity and gender related differences in extended intraesophageal pH monitoring parameters in infants: a retrospective study

Authors:

Dena Nazer (nazer@wayne.edu)
Ronald Thomas (rthomas@crcm.med.wayne.edu)
Vasundhara Tolia (vtolia@med.wayne.edu)

Version: 7 Date: 29 June 2005

Author's response to reviews:

1. Carman and not Carmen for Dept of Peds name.
2. In abstract, in results, comma added after 569 infants.
3. On page 4, line 10 from below, added that PPIs, H2 receptor antagonists, antacids and prokinetics were withheld for 72 hours